Author's response to reviews

Title: No evidence for a decreased risk of thyroid cancer in association with use of metformin or other antidiabetic drugs: a case-control study

Authors:

Claudia Becker (Claudia.Becker@usb.ch)
Susan S Jick (Sjick@bu.edu)
Christoph R Meier (Christoph.Meier@usb.ch)
Michael Bodmer (mi.bodmer@unibas.ch)

Version: 5 Date: 10 July 2015

Author's response to reviews:

Dear Dr Inoue,

Please find enclosed the revised version of the manuscript “No evidence for a decreased risk of thyroid cancer in association with use of metformin or other antidiabetic drugs: a case-control study” which we would like to re-submit to the BMC Cancer.

We modified the relevant sections as requested by the editor:
1. Abbreviations section
2. Competing interests
3. Funding in Acknowledgements
4. Consent statement - whether it was required or not for this study
5. Conclusion heading.

We hereby re-submit the updated version of the manuscript with all changes highlighted.

We hope the manuscript is now acceptable and we would be pleased to see our study published in BMC Cancer.

Sincerely,

Christoph R. Meier, PhD, MSc (corresponding author) on behalf of all authors